Author’s response to reviews

Title: "Reduced fetal growth velocities and the association with neonatal outcomes in appropriate-for-gestational-age neonates: a retrospective cohort study"

Authors:

Manouk Hendrix (manouk.hendrix@mumc.nl; manouk.hendrix@maastrichtuniversity.nl)
Sander van Kuijk (sander.vankuijk@mumc.nl)
Antonio Gavilanes (danilo.gavilanes@mumc.nl)
Desiree Kramer (desiree.kramer@mumc.nl)
Marc Spaanderman (marc.spaanderman@mumc.nl)
Salwan Al-Nasiry (salwan.alnasiry@mumc.nl)

Version: 1 Date: 18 Jan 2018

Author’s response to reviews:

1) Please confirm in your ethics statement whether the participant consent was written or verbal. If it was verbal, please confirm that you received ethics committee approval for verbal consent and detail how you documented the

If the need for informed consent has been waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the Board or a reference to the relevant legislation.

I have explained in the manuscript the ethics statement with informed consent.